Author's response to reviews

Title: Does the local food environment around schools affect diet? Cross-sectional and longitudinal associations in adolescents attending secondary schools in East London

Authors:

Dianna M Smith (dianna.smith@imperial.ac.uk)
Steven Cummins (s.c.j.cummins@qmul.ac.uk)
Charlotte Clark (c.clark@qmul.ac.uk)
Stephen Stansfeld (s.a.stansfeld@qmul.ac.uk)

Version: 2 Date: 15 February 2012

Author's response to reviews:

The following statement is added to the Methods section:

'Ethical approval was obtained from the East London and City Local Research Ethics Committee. All children provided informed written consent. Parents and guardians had an opt-out process whereby they only returned a signed form if they did not want their child to take part, providing informed consent.'